Author's response to reviews

Title: Mannan adjuvants intranasally administered inactivated influenza virus in mice rendering low doses inductive of strong serum IgG and IgA in the lung

Authors:

Owen Proudfoot (oproudfoot@gmail.com)
Sandra Esparon (sesparon@burnet.edu.au)
Choon-Kit Tang (cktang@ifrec.osaka-u.ac.jp)
Karen Laurie (karen.laurie@influenzacentre.org)
Ian Barr (ian.Barr@influenzacentre.org)
Geoff Pietersz (gpietersz@burnet.edu.au)

Version: 2
Date: 22 August 2014

Author's response to reviews:

Philippa Harris
Editor-in-Chief
BMC Infectious Diseases

Dear Dr Philippa Harris,

Please be advise that I have resubmitted the recently submitted manuscript entitled “Mannan adjuvants intranasally administered inactivated influenza virus in mice rendering low doses inductive of strong serum IgG and IgA in the lung”, with the requested revisions made, as follows:

1. I have added line numbers.

2. I have added a “conflicting interests” section/statement.

Thank you for your consideration, and I look forward to hearing from you.

Sincerely,

Owen Proudfoot

Bio-organic and Medicinal Chemistry Laboratory,
Centre for Immunology, Burnet Institute,
85 Commercial Road, Melbourne, 3004,
Tel: 613-9282-2111
Email: owen.proudfoot@burnet.edu.au